Author's response to reviews

Title: Histopathological grading of pediatric ependymoma: reproducibility and clinical relevance in European trial cohorts

Authors:

David W Ellison (David.Ellison@stjude.org)
Mehmet Kocak (Mehmet.Kocak@stjude.org)
Dominique Figarella-Branger (dominique.figarella-branger@univmed.fr)
Felice Giangaspero (felice.giangaspero@uniroma1.it)
Catherine Godfraind (catherine.godfraind@uclouvain.be)
Torsten Pietsch (t.pietsch@uni-bonn.de)
Didier Frappaz (FRAPPAZ@lyon.fnclcc.fr)
Maura Massimino (maura.massimino@istitutotumori.mi.it)
Jacques Grill (Jacques.Grill@igr.fr)
James M Boyett (James.Boyett@stjude.org)
Richard G Grundy (Richard.Grundy@nottingham.ac.uk)

Version: 2 Date: 21 May 2011

Author's response to reviews: see over
Thank you for your advice on the changes required at this, the proof-reading, stage.

I have made all of the major and minor corrections as recommended below.

David Ellison

Formatting changes to be made:
---------------------------------
Major revisions (we require the author to make these changes)

Ethical approval: Any experimental research that is reported in the manuscript should have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Competing interests: Please include a `Competing interests¿ section between the Conclusions and Authors¿ contributions. If there are none to declare, please write `The authors declare that they have no competing interests¿. Please see here for a full list of questions to consider when writing your competing interests statement-
http://www.biomedcentral.com/bmcmed/ifora/?txt_jou_id=2999&txt_mst_id=50733#interests

Authors¿ contributions: Please place an Authors¿ contributions section towards the end of the manuscript text, directly preceding the Acknowledgements. Please use the following format as an example, including using initials to refer to each author¿s contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript. Please use initials instead of full name. Please ensure this is formatted into a paragraph rather than a list or in bullet points.

Additional files: Additional files should not contain copies of the manuscript. These should be removed from the system before acceptance. In order to remove the additional files you will need to make a revised submission.

Figures: the figures have been submitted as more than one page; there are nine figure legends and nine figures cited in the main text, but 18 figure files have been uploaded.
We ask that if a figure consists of separate pages, then a single composite illustration file should be submitted which contains all parts of the figure on a single page. If this is not possible, we recommend that page 1 be submitted as figure 1, page 2 as figure 2, and page 3 as figure 3, etc. Please remember to change any references to the figures in the manuscript file after any changes.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Tables- please remove the visible vertical lines from your tables

Please remove running title from title page.

Please place 'Materials and Methods' to directly after 'Background'.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.